Author's response to reviews

Title: The D9N, N291S and S447X variants in the lipoprotein lipase (LPL) gene are not associated with type III hyperlipidemia

Authors:

David Evans (evans@uke.uni-hamburg.de)
Frank U Beil (beil@uke.uni-hamburg.de)

Version: 4 Date: 22 August 2007

Author's response to reviews:

The corrections suggested by Referee 2 and the formatting requirements requested have been incorporated into the manuscript.